Author’s response to reviews

Title: SET domain containing protein 5 (SETD5) enhances tumor cell invasion and is associated with a poor prognosis in non-small cell lung cancer patients

Authors:

Hairu Yu (1420832604@qq.com)
Jiayi Sun (864922483@qq.com)
Congxuan Zhao (epestion0908@163.com)
Haotian Wang (1370205189@qq.com)
Yeqiu Liu (420447417@qq.com)
Jiajia Xiong (XJJ306@163.com)
Jing Chang (changjing1212@163.com)
Mixue Wang (18334721368@163.com)
Wenhui Wang (609431816@qq.com)
Dongman Ye (cmu-ydm@163.com)
Hongyan Zhou (984274784@qq.com)
Tao Yu (yutao@cancerhosp-ln-cmu.com)

Version: 3 Date: 20 Jun 2019

Author’s response to reviews:

BCAN-D-18-01443

Dear editor

Thanks very much for your reply of our paper titled “SET domain containing protein 5 (SETD5) enhances tumor cell invasion and is associated with the poor prognosis of non-small cell lung cancer”. We have checked carefully.
If my dear editor think this paper needs further polishing, please recommend an official organization to us.

If there is anything else we should do, please do not hesitate to let us know.

Thank you.

Your sincerely

Tao Yu

Department of Medical Imaging, Cancer Hospital of China Medical University, No.44 Xiaoheyan Road, Dadong District, Shenyang, Liaoning Province, China

Department of Medical Imaging, Liaoning Cancer Hospital & Institute, No.44 Xiaoheyan Road, Dadong District, Shenyang, Liaoning Province, China